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D0 Shutdown

* The highest priority activity during the short shutdown was to
upgrade 68 adapter cards for the Silicon to improve the
performance of HDIs recovered during previous shutdown.

® Our secondary goal was to open the forward muon north B-layer

pixel plane to disable a broken MDT wire.

* In addition, routine maintenance and individual channel repairs

have been performed.

— Recover from Power outages

— Repair DCW leak.

13 October 2008 AEM Bill Lee



Do Power Outages

* Several complications resulted from the power outages.

— Monday

* Fire Alarm in MCH1 dumped halon. The alarm was likely triggered
by the heater in the AC unit.

* Other failures: Online switch, CTT Crate controller, L1Cal power
supply, Forward muon crate computer, Rack monitor, online disk
power supply, monitor, and a camera.

— Friday

* Cal preamp power supply, L1 muon card, and a TV monitor.
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w Infrastructure

* Repaired a water leak 1n chilled water system

* Additional cryo system interventions

— Cleared obstruction in heat exchanger.

* Upon opening up the north Calorimeter end cap, we found the
teflon coating on the EC beam pipe
was flaking off.

— Thanks to PPD ES&H for their
help 1n cleaning the coating off
of the beam pipe.
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Data Taking

Delivered Recorded Efficiency Comment
Lum (pb™) Lum (pb™) (%)
12 Oct 1.44 1.16 80

* Silicon Track Trigger rack tripped off several times on
low air tflow.

— The blower was replaced this morning.

— Total downtime was under an hour.
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